CrossMark

GOPEN ACCESS

Citation: Nackers F, Mueller YK, Salih N, Elhag MS, Elbadawi ME, Hammam O, et al. (2015) Determinants of Visceral Leishmaniasis: A Case-Control Study in Gedaref State, Sudan. PLoS Negl Trop Dis 9(11): e0004187. doi:10.1371/journal. pntd.0004187

Editor: Elodie Ghedin, New York University, UNITED STATES

Received: June 18, 2015

Accepted: October 2, 2015

Published: November 6, 2015

Copyright: © 2015 Nackers et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data set underlying the findings of this study is available on request, in accordance with the legal framework set forth by Médecins Sans Frontières (MSF) data sharing policy (Karunakara U, PLoS Med 2013). The MSF data sharing policy ensures that data will be available upon request to interested researchers while addressing all security, legal, and ethical concerns. All readers may contact Ms. Aminata Ndiaye (aminata.ndiaye@epicentre.msf.org) to request the data. RESEARCH ARTICLE

Determinants of Visceral Leishmaniasis: A Case-Control Study in Gedaref State, Sudan

Fabienne Nackers^{1©}*, Yolanda Kathrin Mueller^{1©}, Niven Salih², Mousab Siddig Elhag³, Mobarak Elnour Elbadawi⁴, Omer Hammam², Ann Mumina², Atia Abdalla Atia², Jean-François Etard^{1,5}, Koert Ritmeijer⁶, François Chappuis^{7,8}

Epicentre, Paris, France, 2 Médecins Sans Frontières, Operational Centre Geneva, Khartoum, Sudan,
Federal Ministry of Health, Neglected Tropical Diseases Division, Khartoum, Sudan, 4 State Ministry of
Health, Gedaref, Sudan, 5 Institut de Recherche pour le Développement, Montpellier University, Montpellier,
France, 6 Médecins Sans Frontières, Operational Centre Amsterdam, Amsterdam, the Netherlands,
7 Médecins Sans Frontières, Operational Centre Geneva, Geneva, Switzerland, 8 Geneva University
Hospital, Department of Community Medicine, Geneva, Switzerland

• These authors contributed equally to this work.

* Fabienne.Nackers@Brussels.msf.org

Abstract

Background

Improving knowledge on local determinants of visceral leishmaniasis (VL) is crucial to guide the development of relevant control strategies. This study aimed to identify individual and household level determinants of primary VL in 24 highly endemic villages of Tabarak Allah hospital's catchment area, Gedaref State, Sudan.

Methods

From September 2012 to July 2013, in an unmatched case-control design, 198 patients with primary VL were compared to 801 controls free of VL symptoms and with a negative VL rapid test. Using random spatial sampling, controls were selected with a distribution of age, sex and village of residence proportionate to the distribution of the target population. Data were collected using a structured questionnaire.

Results

Children and men were at higher risk of VL. Reporting VL patient(s) in the household in the previous year was the strongest VL risk factor. In a multivariate analysis, VL risk increased with household size, sleep location (outside the yard, not in the farm), evening outdoor activities in the rainy season (playing, watching TV, radio listening), use of ground nut oil as animal repellent and of smoke of Acacia seyal as indoor repellent, presence of dogs in the yard at night, Acacia nilotica in the yard's immediate surroundings and of a forest at eye range. VL risk appeared to decrease with the use of drinking water sources other than the village water tank, a buffer distance from the adjacent house yard, and with the presence of animals other than dogs in the yard at night. In contrast with previous studies, housing factors,

Funding: This study was funded by Médecins Sans Frontières, Switzerland (MSF-OCG). Some authors are part of MSF-OCG. They participated in the study implementation, interpretation of the results, data collection and revision the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

mosquito-net use, black cotton soil, ethnicity, socioeconomic index, presence of Balanites aegyptica and Azadirachta indica in the yard were not independent VL determinants.

Discussion and conclusion

Although these results do not provide evidence of causality, they provide useful suggestions for guiding further intervention studies on VL preventive measures.

Author Summary

Visceral leishmaniasis (VL), a fatal disease without treatment, is caused by a parasite (*Leishmania*) transmitted to humans through sandflies. Its epidemiology and the vectors involved in its transmission differ importantly between endemic zones. Preventive measures aim at reducing the parasite reservoirs and at limiting human exposure to sandflies. In Sudan, there is a lack of knowledge on VL risk and protective factors, limiting the possibility of development of control strategies. We conducted a study in a VL endemic area, Gedaref State (Sudan), to identify individual and household level determinants of clinical VL. We compared several characteristics of VL patients and inhabitants free of VL. We found that having a household member sick with VL in the previous year was the strongest VL risk factor. VL risk also seemed to be influenced by the household size, sleep location, evening outdoor activities, use of some repellents, and proximity to dogs at night. VL risk also appeared to decrease with a buffer distance between adjacent house yard and with the proximity to animals other than dogs at night. Our findings provide useful suggestions for guiding further intervention studies on VL preventive measures.

Introduction

Visceral leishmaniasis (VL) is a devastating illness with a high case fatality rate when left untreated. Gedaref state is the main endemic area of VL in Sudan where the detection figures vary largely between villages in relation with average rainfall and altitude [1]. Villages with higher incidence are clustered along the Atbarah and Rahad rivers [1], near the Ethiopian border. Since 2009, Médecins Sans Frontières (MSF) is offering diagnosis and treatment for VL patients in Tabarak Allah, near the Atbarah river (Fig 1).

In Eastern Sudan, VL is caused by the parasite *Leishmania donovani* and transmitted through the sandfly *Phlebotomus orientalis* [2,3]. *P. orientalis* sandfly populations tend to peak at the beginning of the rainy season [2,3] and are concentrated in areas with high densitiy of *Acacia seyal* (locally named "Taleh") and *Balanites aegyptica* trees ("Lalob" or "Higleeg") that grow on vertisols (black cotton soil) [4]. *P. orientalis* is believed to bite mainly outdoors, in the evening or in the early morning [4], though some populations of the vector may be more adapted to indoor biting [5]. Although some animals, especially dogs [6], have been shown infected by *L. donovani* in eastern Africa, their role in transmission is unclear and the disease is believed to be mainly anthroponotic [7,8]. In Sudan, post-kala-azar dermal leishmaniasis (PKDL) cases, corresponding to some 50% of treated cases, might act as a reservoir for parasites and play a role in human-to-human transmission [9,10]. Host factors increasing susceptibility to VL include malnutrition and HIV [11].

In the absence of a vaccine, VL preventive measures aim at reducing the parasite reservoirs (human and, where relevant, animal) and at limiting human exposure to sandflies notably





Fig 1. Study area, case-control study on determinants of visceral leishmaniasis, Gedaref State, Sudan, 2012–2013. Left map: Annual incidence of visceral leishmaniasis (VL) in Sudan (2011) from < 5 cases /100,000 inhabitants (yellow) to \geq 15 cases /100,000 inhabitants (red) (Source: Annual Report, Federal Ministry of Health, Sudan). Right map: Annual incidence of VL in Sudan (2011) proportionate to the size and colour of the circles (from 0 to \geq 20 cases /1,000 inhabitants). The black circle includes the 24 study villages.

doi:10.1371/journal.pntd.0004187.g001

through vector control [12]. Risk factors for VL transmission have been studied in Asia, where poverty and housing conditions appeared to consistently influence the risk of VL [13,14]. However, the important continental differences in terms of reservoir, ecology, parasites and vectors involved in VL transmission limit the generalizability of these results to eastern Africa. In Ethiopia [15–17], Kenya [18] and Uganda [19], factors such as proximity to dogs, sleeping outside, under an acacia tree, or in a thatch house, and a low socio-economic status were reported as possible risk factors for clinical VL. In Sudan, younger age, male gender, and ethnicity [20,21] appeared associated with an increased risk of VL while *Azadirachta indica* (locally named "Neem") [21] and use of bednets [15,22] appeared offering some protection. To date in Sudan, VL transmission and its possible individual VL risk factors remain poorly understood. However, further knowledge of local VL determinants is essential to design appropriate control activities. Therefore, we conducted a case-control study in endemic villages of Tabarak Allah area in order to identify individual and household level determinants of primary VL.

Materials and Methods

Study setting and target population

The study was conducted in the catchment area of the MSF Tabarak Allah Hospital, in Al-Gureisha locality, the main provider of VL treatment in the area. The study target area included the 24 villages (out of 45 villages) most affected by VL according to a survey on burden of VL undertaken in 2011 [23], and closest to Tabarak Allah (Fig 1). The population of the study area was estimated at 46,564 individuals in June 2011, with a VL incidence rate estimated at 9.8/ 1000 person-years from June 2010 to June 2011 [23]. The vegetation consists in dry savannah woodland (with *Acacia seyal* and *Balanites aegyptica* trees) combined with black cotton soil. The rainy season usually lasts from June to October.

Study design and sample size

This was an unmatched case-control study. For practical reasons, we expected to enrol cases and controls over a single dry season. Based on the VL incidence rate estimated from June 2010 to June 2011 [23] and accounting for patients that would be missed by the hospital-based passive recruitment, we expected to recruit a bit less than 300 cases. A sample size of 270 cases and 810 controls allowed to detect as significant an odds ratio (OR) of 1.6 with an expected 20% of exposure among controls and a power of 80% (α : 0.05, two-sided).

Sampling procedures

Cases were recruited in Tabarak Allah Hospital. All successive patients presenting clinical signs of VL (*ie* history of fever of two weeks or more; and either splenomegaly, lymphadenopathies or wasting) and who tested positive to a rk39-based rapid test (BioRad IT-Leish) [24] and/or a direct agglutination test (DAT; titre \geq 1:6400) [25,26] and/or had a positive lymph node aspirate [27] were eligible for recruitment if they were at least 6 month-old and if they had been resident of a village of the target area since at least one year. Patients previously treated for VL were excluded. All eligible cases (or their caregivers for individuals younger than 15 years) were invited to sign a written informed consent for participating in a structured interview during a home visit in the month following discharge. HIV-co-infection was not systematically assessed but was expected to be low [28].

Controls were selected from the population with a distribution of age, sex and villages of residence proportionate to the distribution observed in a survey in 2011 [23]. Control houses were selected randomly using a geographic sampling procedure. The perimeter of the inhabited area of each village was recorded in the field using global positioning system (GPS) devices and saved as shapefiles. In each village perimeter, a pre-specified number of GPS random points, according to the 2011 population distribution, were generated. Control selection started in the nearest house from each point. If several houses were at the same distance, one house was selected at random (using a random number table). In each selected house, one individual of a pre-specified sex and age category (to fit the 2011 distribution) living in this house since at least one year, free of clinical symptoms of VL (including PKDL) and with no history of previous treatment for VL was randomly selected. After written informed consent, each potential control underwent a BioRad IT-Leish test. Individuals with a positive result were excluded. If the rapid test was negative, a structured interview was conducted. If there was no eligible control in the first house, the control selection was continued in the nearest one. In case of absence, a second visit was planned within the week. The recruitment started with the dry season on the 15th of September 2012 and ended with the beginning of the rainy season on the 15th of June 2013. We aimed to ensure a similar timing of recruitment for cases and controls. Interviews of controls took place between the 22nd September 2012 and the 9th of July 2013 and all villages were regularly visited over the all study period.

Data collection

Cases and controls were invited to answer a structured questionnaire on possible determinants of VL (<u>S1 File</u>). It was designed for an exploratory approach and it included data on demographic characteristics, socio-economic factors, personal medical history, travel history, housing conditions, daily activities (with a focus on evening activities), description of house yard and immediate surroundings including animal presence, and occurrence of VL in the house or the adjacent house. When relevant, different answers were allowed for the rainy and the dry seasons. The cases and controls were asked to describe their usual behaviour during the current dry season (for cases, before they developed VL symptoms) and the last rainy season. Questionnaires were translated in Arabic and were piloted before the start of the interviews. The interviewers were trained to administer the questionnaires in a standardised way.

Data analysis

Data were double-entered and analysed using Stata 12 software (Stata Corporation, College Station, Texas, USA). Using a principal component analysis, a socioeconomic index including data of ownership among controls (radio, TV, bicycle, motorcycle, car/truck/tractor, mobile phone, donkey cart, and generator) was built. Each variable in the index was given a coefficient corresponding to eigenvectors of the first and second components, weighted for their respective contribution. The index was divided into quintiles.

Variables were grouped by predefined thematic sections (demographics, travel history, activities and sleeping habits, characteristics of the household, the house, house yard and surroundings) and by season. The thematic sections are presented in the supplementary material (S1 Table), where individual level and household level determinants are distinguished. In each section, the variables associated with VL (p<0.20) after adjustment for age, sex and village taken as a random effect (due to the high number of villages) were combined in mixed multivariate logistic regression models using a stepwise forward approach. Variables associated with VL (p<0.20) from the different sections were then combined to build one multivariate model with potential VL risk factors at the individual level and another multivariate model with household-level risk factors. Variables not associated with VL (p \ge 20%) were excluded. Finally all variables were combined into a multivariate logistic model in which variables significant at the 5% level were kept. An interaction between age and sex (likelihood ratio test for interaction p = 0.027) was kept in the model. The variables related to the presence of VL patients in the household or the adjacent house(s) were kept separate because they were likely on the causal pathway between VL and the other determinants.

Ethics statement

Authorization to conduct the study was obtained from the Gedaref Ministry of Health, Humanitarian Affairs Commission, and the village authorities. Ethical clearance was obtained from the MSF ethical review board and National Research Ethics Review Committee of the 'Health Research Council" (National Ministry of Health). All adult subjects provided informed consent, and a parent or guardian of any child participant provided informed consent on their behalf. Each individual signing the consent form to participate as a control was offered a mosquito net. Individuals with VL clinical symptoms were referred to Tabarak Allah for further diagnostic. Because of the possibility of a not yet clinical infection, asymptomatic individuals with a positive rk39 test were followed-up by phone during 6 months and encouraged to consult Tabarak Allah hospital in case they developed persistent fever in the months following the study.

Results

Of the 321 primary VL patients admitted in Tabarak Allah during the study period, 208 were coming from the study area, corresponding to a passive detection rate of 6.0/1000 person-years (ranging from 0 to 16.4 person-years across villages). Nine did not live in the same place during the past year and one patient died, resulting in 198 cases interviewed. Of the 849 eligible individuals selected as controls, 3 refused participation, 21 were absents despite two revisits, nine presented VL clinical symptoms and 15 were rK39 positive (of which 4 developed VL during the follow-up, 5 remained asymptomatic, and 6 were lost to follow-up). In total, 801 controls were included in the analysis.

	Control (N =	801)	Case (N = 19		
	n	%	n	%	p-value
Age (years)					
0 to 9	283	35.3	104	52.5	<0.001
10 to 19	196	24.5	52	26.3	
20 to 39	187	23.4	24	12.1	
40 or more	135	16.9	18	9.1	
Male	414	51.7	124	62.6	0.006
Ethnicity (60 missing)					
Massalit	263	34.8	57	31.0	<0.001
Zabarma/Fallata/Hausa	198	26.2	24	13.0	
Tama	52	6.9	37	20.1	
Arnga/Gemir	63	8.3	19	10.3	
Sudanese, Arabs	81	10.7	14	7.6	
Sudanese, Non Arabs	52	6.9	22	12.0	
Others	46	6.1	11	6.0	
Village					
Berber Al-Fugera	87	10.9	59	29.8	<0.001
Tabarak Allah	84	10.5	43	21.7	
Khuor Zaraf	78	9.7	7	3.5	
Jebel Ghana	63	7.9	24	12.1	
Al-asira	62	7.7	10	5.1	
Um Gzaz	54	6.7	1	0.5	
Birkat Norein	52	6.5	11	5.6	
Mashra Al Forsan	51	6.4	8	4.0	
Wad Arood	41	5.1	0	0.0	
Other	229	28.6	35	17.6	
Socioeconomic index					
Lowest	204	25.5	46	23.2	0.475
2nd lowest	255	31.8	54	27.3	
Middle	34	4.24	8	4.0	
2nd highest	185	23.1	52	26.3	
Highest	123	15.4	38	19.2	

Table 1. Socio-demographic characteristics, case-control study on determinants of visceral leishmaniasis, Gedaref State, Sudan, 2012–2013.

doi:10.1371/journal.pntd.0004187.t001

Considering socio-demographic characteristics (<u>Table 1</u>), cases were younger than controls and more often male. A third of cases and controls belonged to the Massalit, other major ethnic groups being Zabarma, Fallata, Hausa (26.2% of controls) and Tama (20.1% of cases). Cases originated mainly from Berber Al-Fugera (29.8%) and Tabarak Allah (21.7%), compared to 21.4% of the controls. There was no evidence for a difference of socio-economic index category between cases and controls. Univariate analysis comparing all cases and controls characteristics is presented in the supplementary material (<u>S1 Table</u>).

After adjustment on age, sex, their interaction and village as random effect, there was no evidence ($p \ge 0.20$) for an association between VL and black cotton soil (surrounding the yard of 85.4% of the cases and 74.7% of the controls), proximity of water bodies, housing materials, use of mosquito nets, indoor spraying, travel history, day activities, and long term medical treatment. All variables that remained associated with VL (p < 0.20) after adjustment are presented in the <u>Table 2</u> with OR and 95% confidence intervals (95%CI). Among participants of

schooling age or older, cases were more frequently attending school. Between sunset and bedtime, the majority of the participants stayed exclusively outdoor in the dry season (S1 Table) while, in the rainy season, 30.4% of controls and 26.8% of cases stayed indoor only (Table 2). In the rainy season, outdoor evening activities (playing, watching TV/radio listening) were associated with an increased risk of VL compared to staying indoor. Most participants reported to sleep within the house yard, although 2.9% of controls and 7.6% of cases reported to sleep outside the house yard but within the village (for example, some children slept at school). During the rainy season, 15.5% of controls and 8.6% of cases reported to sometimes sleep in the field/farm outside the village. In the dry season, about half of the participants were sleeping outside a room while, in the rainy season, more than 90% of all participants were sleeping inside a room. Cases belonged to bigger households than controls and tended to share the sleeping room with more people than controls. Presence of cracks in the room's wall (made of either strong mud, plastered or unplastered cane) tended to increase the risk of VL. Water sources were concordant in the dry and the rainy season (Cohen's kappa 0.75) and differed between cases and controls, the main source of drinking water being surface water for controls while cases tended to drink more often water from the village water tank. Trees tended to be more frequent in and around cases yards but, after adjustment, only the presence of Azadirachta indica and Acacia nilotica appeared more likely associated (p<0.05) with an increased risk of VL. Increasing distance between the house yard and the adjacent yard tended to reduce the risk of VL while the presence of a forest at eye range from the house appeared to increase the risk of VL. At night, in both seasons, animals (donkeys, goats, sheep, cattle and dogs) were reported in the majority of participants' yards. The presence of dogs in the yard at night was associated with an increased risk of VL while the presence of other animals in the yard appeared rather associated with a reduced VL risk.

The use of smoke from *Acacia seyal* as indoor repellent in the rainy season was associated with an increased risk of VL. Also, repellent for animal (chemical, local tar, smoke from grass, natural oil) was frequently used but only the use of ground nut body oil was associated with an increased risk of VL.

The final multivariate model combining independent individual and household VL determinants is presented in the <u>Table 3</u>.

As shown in the <u>Table 4</u>, during the year prior the study, 7.4% of controls compared to 71.7% of cases had at least one household member who suffered from VL (p<0.001) and, among those, cases had more often 2 to 4 household members affected (21.8% versus 10.2%). Before that, VL was reported to have occurred in the households of about 74% of all participants. In the year prior the study, a patient with VL in an adjacent house was reported by 30.8% of the controls and in 42.6% of the cases. Before that period, VL was reported to have occurred in a neighbour of 95.6% of the controls and 90.2% of the cases. Adjusting for individual and household determinants did not affect the size of the associations between VL and previous VL patients in the household or an adjacent house (<u>Table 4</u>). When combining the variables related to VL patients in the household and the adjacent houses in a multivariate model, reporting a VL patient(s) in the household in the previous year remained the only independent VL determinant.

Discussion

In this study, living in the proximity of VL patients was the strongest VL risk factor. Three out of four cases reported having a household member sick with VL in the last year compared to less than one out of ten controls. This has also been reported in the Indian subcontinent [13,29] and for asymptomatic VL in Ethiopia [30]. It underlines the importance of studying

Table 2. Individual level determinants associated (p<0.20) with VL after adjustment: crude and adjusted analysis. Gedaref State, Sudan, 2012–2013.

	Control (N = 801)		Case) (N = 198)		Univariate				Adjusted*			
	n	%	n	%	Crude OR	95%	S CI	LR test	Adjusted OR	95%	% CI	LR test
Gender, in <10 years												
Male	146	51.8	69	66.4	1			0.005	1			<0.001
Female	136	48.2	35	33.7	0.46	0.29	0.74		0.41	0.25	0.68	
Gender, per 10 years												
Male	267	51.5	55	58.5	1				1			
Female	251	48.5	39	41.5	0.56	0.4	0.8		0.53	0.36	0.76	
Age, per 10 years, among males												
Median (IQR)	14	(7– 35)	8	(6– 14)	0.6	0.44	0.83		0.55	0.39	0.77	
Age, per 10 years, among females												
Median (IQR)	14	(7– 30)	10	(6– 24)	0.73	0.64	0.85		0.7	0.6	0.81	
Education (281 too young for school excluded)												
Illiterate	126	21.4	18	13.9	1			<0.001	1			<0.001
Primary school	122	20.7	15	11.6	0.86	0.41	1.78		0.7	0.3	1.64	
Secondary school	35	5.9	6	4.6	1.2	0.44	3.25		0.8	0.26	2.49	
Literate via Koranic school	144	24.4	12	9.3	0.58	0.27	1.26		0.66	0.27	1.59	
Currently attending school	162	27.5	78	60.5	3.37	1.92	5.92		2.57	1.17	5.68	
Occupation												
No occupation**	586	73.1	172	86.7	1			<0.001	1			0.166
Farming/herding	149	18.6	17	8.7	0.39	0.23	0.66		0.53	0.26	1.1	
Trade /skilled labour	66	8.2	9	4.5	0.46	0.23	0.95		0.54	0.22	1.31	
History of travel in Gedaref State (outside Gureisha/Gedaref town)	38	4.7	5	2.5	0.52	0.2	1.34	0.134	0.45	0.17	1.22	0.093
Bedtime hour*** (20 missing)												
17h to 20h	237	43	112	57.1	3.09	1.37	6.93	<0.001	1.99	0.82	4.82	0.145
21h to 22h	381	48.7	77	39.3	1.88	0.83	4.25		1.46	0.61	3.48	
23h to 2h	65	8.3	7	3.6	1				1			
Place of stay from sunset to sleep*** (50 in bed before/ at sunset excluded)												
Indoor only	213	27.9	44	23.5	1			0.057	1			0.038
Outdoor	314	41.2	67	35.8	1.07	0.71	1.62		1.17	0.74	1.84	
Both indoor and outdoor	235	30.8	76	40.6	1.55	1.03	2.35		1.74	1.1	2.73	
Main evening outdoor activities *** (1 missing)												
Stay indoor only	243	30.4	53	26.8	1			<0.001	1			<0.001
Playing	154	19.3	65	32.8	1.94	1.28	2.94		1.71	1.08	2.7	
TV / radio	52	6.5	27	13.6	2.39	1.38	4.15		2.67	1.42	5.03	
Discussing-relaxing	265	33.1	47	23.7	0.82	0.53	1.25		1.02	0.62	1.66	
Cooking/house activities	47	5.9	4	2	0.39	0.14	1.13		0.52	0.17	1.59	
Other activities	39	4.9	2	1	0.24	0.05	1.01		0.31	0.07	1.4	
Sleep location***												
Outside the house yard	23	2.9	15	7.6	2.77	1.42	5.42	0.004	3.33	1.52	7.31	0.003
Never sleeping in the field/farm***	676	84.5	181	91.4	1.95	1.14	3.32	0.009	2.12	1.16	3.91	0.012
Increase per 1 household member	_							_				
Mean household size (SD)	7.7	-3.3	8.4	-4	1.06	1.01	1.1	0.012	1.06	1.01	1.12	0.024

(Continued)

Table 2. (Continued)

	Control (N = 801)		Case (N = 198)		Univariate			Adjusted*				
	n	%	n	%	Crude OR	95%	% CI	LR test	Adjusted OR	959	% CI	LR test
Gender of the head of household												
Female	38	4.7	5	2.5	0.53	0.2	1.36	0.153	0.46	0.17	1.28	0.113
Nb sleeping in room***, per 1 person												
Median (IQR)	4	(3–6)	5	(3–6)	1.06	1.01	1.12	0.002	1.04	1	1.09	0.023
Walls cracked (1 missing)												
No	393	49.1	82	41.4	1			0.013	1			0.079
Yes, but not much	226	28.3	51	25.8	1.05	0.71	1.54		0.9	0.59	1.38	
Yes, many cracks	181	22.6	65	32.8	1.72	1.19	2.49		1.49	0.98	2.25	
Termites in the room (1 Missing)									1			
No	432	54	118	59.6	1			0.065				0.076
Yes, but not so many	203	25.4	35	17.7	0.63	0.42	0.95		0.62	0.4	0.96	
Yes, many	165	20.6	45	22.7	1	0.68	1.47		1	0.65	1.54	
Source of drinking water (in the dry season) (1 missing)									1			
Village water tank	162	20.2	73	36.9	1			<0.001				0.064
River/surface water	342	42.7	41	20.7	0.27	0.17	0.41		0.44	0.22	0.86	
Pump, well, water sellers	297	37.1	84	42.4	0.63	0.43	0.91		0.75	0.5	1.14	
Termite hills in house yard												
Yes (versus no)	94	11.7	11	5.6	0.44	0.23	0.84	0.007	0.59	0.29	1.17	0.113
Shelter in the house yard												
Yes (versus no)	717	89.5	170	85.9	0.71	0.45	1.13	0.155	0.66	0.4	1.1	0.119
Animal accommodation in the yard												
Yes (versus no)	558	69.7	128	64.7	0.8	0.57	1.11	0.176	0.78	0.54	1.12	0.183
Poultry accommodation in the yard												
Yes (versus no)	605	75.5	131	66.2	0.63	0.45	0.89	0.009	0.75	0.52	1.09	0.135
Forest/wood at eye range												
Yes (versus no)	255	31.9	80	40.4	1.44	1.05	1.99	0.026	1.49	1.02	2.17	0.039
Balanites aegyptica around the yard												
Yes (versus no)	694	86.5	186	93.9	2.6	1.37	4.93	0.001	1.6	0.77	3.34	0.195
Azadirachta indica around the yard												
Yes (versus no)	695	87	181	91.4	1.59	0.93	2.73	0.076	1.92	1.06	3.49	0.026
Acacia nilotica around the yard												
Yes (versus no)	348	43.5	125	63.1	2.22	1.61	3.07	<0.001	1.77	1.24	2.55	0.002
Acacia mellifera around the yard												
Yes (versus no)	183	22.9	71	35.9	1.88	1.35	2.63	<0.001	1.31	0.9	1.91	0.158
Distance to the closest house yard (1 missing)												
Share a common limit	591	73.9	169	85.4	1			0.001	1			0.005
Space in between (<10m)	119	14.9	14	7.1	0.41	0.23	0.74		0.41	0.22	0.76	
More than 10 m	90	11.3	15	7.6	0.58	0.33	1.03		0.62	0.34	1.16	
Animals in the yard at night***												
Yes (<i>versus</i> no)	688	86	158	79.8	0.64	0.43	0.96	0.034	0.69	0.44	1.08	0.107
Sheep in the yard at night***												
Yes (versus no)	296	37	59	29.8	0.72	0.52	1.01	0.057	0.78	0.54	1.12	0.177
Dog in the yard at night***												
Yes (<i>versus</i> no)	55	6.9	21	10.6	1.58	0.93	2.68	0.1	1.92	1.05	3.49	0.037

(Continued)

Table 2. (Continued)

	Control (N = 801)		Case (N = 198)		Univariate			Adjusted*			
	n	%	n	%	Crude OR	95% CI	LR test	Adjusted OR	95% CI	LR test	
Uncovered animal burrows in the yard*** (2 don't know)											
No	325	40.7	75	37.9	1		0.15	1		0.187	
Yes, but not many	71	8.9	11	5.6	0.67	0.34 1.33		0.62	0.3 1.28		
Yes, many	403	50.4	112	56.6	1.2	0.87 1.67		1.15	0.8 1.64		
Fire or smoke as repellent indoor *** (3 missing, 2 don't know)											
Daily	295	36.9	92	47.2	1		0.027	1		0.084	
Frequently / sometimes	193	24.2	43	22.1	0.71	0.48 1.07		0.78	0.5 1.21		
Rarely / never	311	38.9	60	30.8	0.62	0.43 0.89		0.74	0.49 1.11		
Smoke from Acacia seyal as indoor repellent	434	54.2	132	66.7	1.69	1.22 2.34	0.001	1.47	1.02 2.12	0.038	
Ground nut oil as animal body repellent	18	2.3	12	6.1	3.38	1.56 7.34	0.003	3.57	1.49 8.58	0.005	

LR: Likelihood ratio; VL: visceral leishmaniasis; OR: Odds Ratio; 95%CI: 95% Confidence Interval;

* Adjusted for age, sex, their interaction, and village as random effect;

**Housework, too young, attending school, unemployed.

*** In the rainy season as a proxy of the whole year (trend of association being similar in both seasons though some factors less frequent in the dry season).

doi:10.1371/journal.pntd.0004187.t002

further the transmission dynamics between patients and their household contacts [31]. However, the role of PKDL as potential reservoir remains unclear. Indeed, reporting one or more household member(s) sick with VL in the past year was associated with a lower OR if the household member(s) also developed a rash compatible with PKDL. This might suggest that individuals with PKDL are less infectious than individuals with active VL but this might also result from misclassification of rashes as being PKDL although these were due to other diseases.

As previously reported, the risk of VL was higher in males and the young age groups [21] and tended to increase with household size which logically increases the number of possible cases in the household [17]. However, this increased VL risk with the household size might have been overestimated due to the sampling procedures that tended to favour controls from smaller households (by selecting only one control per house). However, important overestimation is unlikely because weighting the final multivariate model to account for the number of eligible controls per house did not change the estimation of the OR nor its precision (S2 Table). By contrast [20,21], ethnicity was not associated with VL after adjustment despite the Fallata/ Hausa/Zabarma ethnic group being less represented among VL cases. This does not support the hypothesis of a genetic predisposition independent from environmental or behavioural factors. Most ethnic groups settled in this VL endemic area a long time ago possibly reducing the difference in terms of exposure and development of natural immunity.

Also in contrast, there was no evidence for an association between VL and neither the socioeconomic [15,19,32] nor housing factors [15,17]. Cracks in the walls, an important factor in Ethiopia [17], increased the risk of VL in adjusted but not in the multivariate analysis. This supports a predominant role of outdoor biting, as does the association with VL risk and outdoor evening activities (such as watching TV/listening to the radio or playing). This was more

Table 3.	Multivariate logistic regression mixed model of individual and household determinants for visceral leishmaniasis,	with village as a ran
dom effe	ect. Gedaref, Sudan, 2012–2013.	

(N = 998)	OR*	95%	95%CI*					
Gender, in <10 years								
Male	1			<0.001				
Female	0.38	0.23	0.65					
Gender , \geq 10 years								
Male	1			<0.001				
Female	0.51	0.34	0.78					
Age, per 10 years, among males	0.53	0.23	0.65	<0.001				
Age, per 10 years, among females	0.71	0.59	0.85	<0.001				
Main evening outdoor activities**								
Stay indoor only	1			<0.001				
Playing	1.77	1.08	2.90					
TV / radio	3.84	1.92	7.68					
Discussing-relaxing	1.44	0.85	2.45					
Cooking/house activities	0.55	0.17	1.83					
Other activities	0.39	0.08	1.97					
Sleep location**								
In the house yard	1			0.002				
Outside the house yard	3.91	1.70	9.00					
Never sleeping in the field/farm**	2.62	1.32	5.17	0.004				
Household size (increase of one person)	1.07	1.01	1.13	0.024				
Source of drinking water (dry season)								
Village water tank	1			0.014				
River/surface water	0.36	0.18	0.71					
Pump, well, water sellers	0.63	0.40	0.99					
Forest at eye range	1.71	1.14	2.55	0.009				
Acacia nilotica in yard's surroundings	1.72	1.17	2.53	0.006				
Distance to the closest house								
Share a common limit	1			<0.001				
Space in between (<10m)	0.35	0.18	0.67					
More than 10 m	0.51	0.26	1.00					
Animals in the yard at night**	0.56	0.34	0.92	0.024				
Dog in the yard at night**	2.43	1.27	4.69	0.010				
Smoke from Acacia seyal as indoor repellent	1.57	1.06	2.32	0.024				
Ground nut oil as animal body repellent	4.15	1.65	10.5	0.003				

OR: Odds Ratio; 95%CI: 95% Confidence Interval;

*Adjusted for village as a random effect (14% of variance, 95%CI 5-35%);

** During the rainy season; The level of education of the participant was not included in the model due to collinearity with age.

doi:10.1371/journal.pntd.0004187.t003

obvious during the rainy than the dry season during which most inhabitants remain unprotected outdoor in the evening. This outdoor exposure during the known biting-time and season of the sandflies might explain why bed-nets did not appear protective against clinical VL, adding to the mixed results reported in the literature [15,33]. Innovative preventive measures deserve further consideration. For example, it might be interesting to evaluate the effectiveness of targeted personal protection strategies, such as the use of adapted chemical repellent use or

Univariate Adjusted* Control Cases (N = 801) (N = 198) Crude OR 95% CI LR test OR* 95% CI* LR test* % % n n In the past year Among household members ** No VL < 0.001 742 92.6 56 28.3 1 < 0.001 1 38.4** > 1 VL case but no or unknown rash 44 5.5 117 59.1 35.2 22.7 54.7 22.8 64.7 \geq 1 VL case who developed a rash 15 1.9 25 12.6 22.1 11.0 44.3 20.1** 8.9 45.4 Among adjacent house (17 missing) 0.003 0.235 No VL 69.2 109 57.4 1 1 548 > 1 VL case but no or unknown rash 188 23.7 56 29.5 1.50 1.04 2.15 1.05 0.68 1.61 2.24 1.68 \geq 1 VL case who developed a rash 56 7.1 25 13.2 1.34 3.75 0.93 3.04 Before past year Among household members (3 missing) No VL 213 26.6 45 23.1 1 0.360 1 0.705 > 1 VL case but no or unknown rash 423 52.8 114 58.5 1.28 0.87 1.87 0.88 0.56 1.4 165 1.03 0.64 0.79 1.4 \geq 1 VL case who developed a rash 20.6 36 18.5 1.67 0.45 Among adjacent households (143 missing) 0.061 0.020 No VI 31 4.4 14 8.8 1 1 \geq 1 VL case but no or unknown rash 422 60.6 99 62.3 0.52 0.27 1.01 0.39 0.17 0.91 \geq 1 VL case who developed a rash 244 35.0 46 28.9 0.42 0.21 0.85 0.29 0.12 0.69

Table 4. Patients with visceral leishmaniasis in the household or an adjacent house: crude and adjusted analysis. Gedaref State, Sudan, 2012–2013.

LR: Likelihood ratio; VL: visceral leishmaniasis; OR: Odds Ratio; 95%CI: 95% Confidence Interval;

* Adjusted for age, sex, their interaction, and village as random effect and for all determinants included in the multivariate model presented in the <u>Table 3</u>; ** Only independent VL determinant found when combining the variables related to VL patients in the household and the adjacent houses in a

multivariate model.

doi:10.1371/journal.pntd.0004187.t004

insecticide-impregnated clothing for household members of VL cases or to inhabitants of villages currently experiencing an increase in VL cases.

Although the most frequent soil around the participants' yards, there was no association with VL and black cotton soil, whose cracks might offer breeding sites to sandflies [34]. The predominance of black cotton soil in the study area might have limited the possibility to find difference between cases and controls and, consequently, an association with VL. Sandy soil in or around the house yard appeared protective in the univariate analysis but not after adjustment. Soil composition is quite homogenous at village level, and in the yard, the soil is often covered with gravels or sand filling the cracks. This might possibly limit sandflies breeding opportunities and might explain sleeping outside the house yard as a VL risk factor. Also, a buffer space between house yards appeared to offer some protection, possibly against transmission from VL cases living in neighbouring yards.

Having a forest, a likely natural focus of sandflies, at eye range from the house yard appeared as a risk factor for VL as well as the presence of *Acacia nilotica* in its immediate surroundings. However, *Balanites aegyptica*, *Acacia seyal* [4,20] and *Azadirachta indica* [21], that have been reported associated with VL, were not in this study after adjustment. Only the use of smoke of *Acacia seyal* as indoor repellent appeared as an independent determinant of VL. During the time of the survey, about a third of the villages investigated were involved in planting *Azadirachta indica* or cutting *Balanites aegyptica and Acacia seyal*. Therefore reverse cause bias

cannot be excluded because such activities might have been more intense around the houses of cases. Also, the surrounding vegetation, and microecological conditions influencing sandflies density, might have been different in the farm and the village. Consequently, sleeping in the field or the farm was associated with a decrease in VL risk, while the opposite was found in Ethiopia [17,35].

The association between having a dog in the yard at night and the risk of VL reinforces the role for dogs in sustaining VL transmission in eastern Africa [16,36]. Asymptomatic dogs have been found carriers of *Leishmania* in Sudan [37,38]. The presence of other animals in the yard at night was found protective. In Ethiopia, *P. orientalis* were shown to feed mainly on cattle [39] although none of the sandflies caught in that study was infected with *Leishmania*. Presence of animals or their ownership has alternatively been reported either as VL risk factor or protective factor [13,14,17,19,21]. Consequently, it is important to study locally the possible zoonotic transmission (especially with dogs) and to analyse the bloodmeal of sandflies caught in the house yards of the area. The use of ground nut oil as an animal repellent was associated with an increased risk of VL. The use of insecticide for livestock has also been reported as a VL risk factor in Kenya [19]. This suggests that ground nut oil either repel sandflies that consequently feed on another unprotected host, or actually attracts sandflies to the area. This finding needs confirmation but suggest that other animal repellents should be explored as alternative to ground nut oil.

The data were collected through a structured questionnaire by surveyors carefully trained to ensure standardization of its administration. However, a questionnaire might not always capture accurately the variation in population behaviours. Surveyors were not blinded of the participant status and cases were asked to recall the time period before the onset of the case VL symptoms. This might have contributed to inaccuracies and possible recall or observer bias. Also, because numerous variables were investigated and multiple statistical tests done, random findings cannot be excluded. Notably, the association observed between VL and the source of drinking water is unclear and might result from multiplicity of statistical testing or residual confounding due to imperfect adjustment on villages of residence.

Cases were recruited in the Tabarak Allah hospital. Patients with VL that did not attend this health facility were not included and this might have limited the comparability of cases to community control. However these likely represented only a small fraction of all VL cases occurring in the area where even habitants living in decentralised villages tend to present to Tabarak Allah hospital spontaneously [23]. Also, patients from nomadic population or seasonal workers were not included and this might limit the generalizability of the results. Although an important population at risk in some area [15], they represented less than 5% of the primary VL cases in Tabarak Allah during the study period.

Finally, this study compared VL clinical cases to non-infected controls. Consequently, it evaluated potential determinants for both infection and progression towards VL clinical disease but cannot disentangle these two aspects. In order to fully evaluate determinants for infection, we should also have included cases with subclinical infection. However, laboratory methods to identify subclinical infections are imperfect, which would have led to misclassification. Similarly, the duration of positive rK39 in subclinical infection is unknown and it is thus likely that individuals with past subclinical infection have been included as controls, possibly leading to an underestimation of the strength of some associations.

In conclusion, because of the exploratory approach, these results do not provide strong evidence of causality but still, they provide useful suggestions for guiding further intervention studies on VL preventive measures. Control strategies would benefit from further research on VL transmission within house yards and across villages. In the meantime, efforts should target households of patients diagnosed with VL.

Supporting Information

S1 File. Questionnaire in English. (PDF)

S1 Table. Univariate analysis of all determinants. Case-control study, Sudan, 2012–13. (DOC)

S2 Table. Multivariate models of individual and household determinants for visceral leishmaniasis (with village as a random effect).

(DOC)

S1 Checklist. STROBE checklist for case-control studies. (DOC)

Acknowledgments

We are thankful to the Ministry of Health of Gedaref, the Federal Ministry of Health of Sudan, and the University of Gedaref for their support and relevant advices. We also thank our colleagues from Epicentre and the MSF-OCG headquarters, the study teams, the MSF field team and the villagers who took part in this study.

Author Contributions

Conceived and designed the experiments: YKM FN FC JFE KR. Performed the experiments: FN NS AAA OH AM. Analyzed the data: YKM. Wrote the paper: FN YKM. Interpretation of the results:FN YKM NS MSE MEE OH AM AAA JFE KR FC. Revision of the manuscript: FN YKM NS MSE MEE OH AM AAA JFE KR FC. Approval of the final manuscript: FN YKM NS MSE MEE OH AM AAA JFE KR FC. Scientific and technical support: MSE MEE.

References

- Elnaiem DE, Schorscher J, Bendall A, Obsomer V, Osman ME, Mekkawi AM, et al. Risk mapping of visceral leishmaniasis: the role of local variation in rainfall and altitude on the presence and incidence of kala-azar in eastern Sudan. Am J Trop Med Hyg 2003 Jan; 68(1):10–7. PMID: <u>12556141</u>
- 2. Elnaiem DA, Hassan HK, Ward RD. Phlebotomine sandflies in a focus of visceral leishmaniasis in a border area of eastern Sudan. Ann Trop Med Parasitol 1997 Apr; 91(3):307–18. PMID: <u>9229023</u>
- Elnaiem DA, Ward RD, Hassan HK, Miles MA, Frame IA. Infection rates of Leishmania donovani in Phlebotomus orientalis from a focus of visceral leishmaniasis in eastern Sudan. Ann Trop Med Parasitol 1998 Mar; 92(2):229–32. PMID: <u>9625920</u>
- Elnaiem DE. Ecology and control of the sand fly vectors of Leishmania donovani in East Africa, with special emphasis on Phlebotomus orientalis. J Vector Ecol 2011 Mar; 36 Suppl 1:S23–S31. doi: <u>10.</u> <u>1111/j.1948-7134.2011.00109.x</u> PMID: <u>21366778</u>
- Lambert M, Dereure J, El-Safi SH, Bucheton B, Dessein A, Boni M, et al. The sandfly fauna in the visceral-leishmaniasis focus of Gedaref, in the Atbara-River area of eastern Sudan. Ann Trop Med Parasitol 2002 Sep; 96(6):631–6. PMID: <u>12396326</u>
- Hassan MM, Osman OF, El-Raba'a FM, Schallig HD, Elnaiem DE. Role of the domestic dog as a reservoir host of Leishmania donovani in eastern Sudan. Parasit Vectors 2009; 2(1):26. doi: <u>10.1186/1756-3305-2-26</u> PMID: <u>19534802</u>
- Zijlstra EE, El-Hassan AM. Leishmaniasis in Sudan. Visceral leishmaniasis. Trans R Soc Trop Med Hyg 2001 Apr; 95 Suppl 1:S27–S58. PMID: <u>11370250</u>
- Ready PD. Epidemiology of visceral leishmaniasis. Clin Epidemiol 2014; 6:147–54. doi: <u>10.2147/</u> CLEP.S44267 PMID: 24833919
- Desjeux P, Ghosh RS, Dhalaria P, Strub-Wourgaft N, Zijlstra EE. Report of the Post Kala-azar Dermal Leishmaniasis (PKDL) Consortium Meeting, New Delhi, India, 27–29 June 2012. Parasit Vectors 2013; 6:196. doi: <u>10.1186/1756-3305-6-196</u> PMID: <u>23819611</u>

- Zijlstra EE, Khalil EA, Kager PA, El-Hassan AM. Post-kala-azar dermal leishmaniasis in the Sudan: clinical presentation and differential diagnosis. Br J Dermatol 2000 Jul; 143(1):136–43. PMID: 10886148
- 11. Leta S, Dao TH, Mesele F, Alemayehu G. Visceral leishmaniasis in Ethiopia: an evolving disease. PLoS Negl Trop Dis 2014 Sep; 8(9):e3131. doi: <u>10.1371/journal.pntd.0003131</u> PMID: <u>25188253</u>
- Stockdale L, Newton R. A review of preventative methods against human leishmaniasis infection. PLoS Negl Trop Dis 2013; 7(6):e2278. doi: <u>10.1371/journal.pntd.0002278</u> PMID: <u>23818997</u>
- Bern C, Courtenay O, Alvar J. Of cattle, sand flies and men: a systematic review of risk factor analyses for South Asian visceral leishmaniasis and implications for elimination. PLoS Negl Trop Dis 2010; 4(2): e599. doi: <u>10.1371/journal.pntd.0000599</u> PMID: <u>20161727</u>
- Singh SP, Hasker E, Picado A, Gidwani K, Malaviya P, Singh RP, et al. Risk factors for visceral leishmaniasis in India: further evidence on the role of domestic animals. Trop Med Int Health 2010 Jul; 15 Suppl 2:29–35. doi: 10.1111/j.1365-3156.2010.02515.x PMID: 20487424
- Argaw D, Mulugeta A, Herrero M, Nombela N, Teklu T, Tefera T, et al. Risk factors for visceral Leishmaniasis among residents and migrants in Kafta-Humera, Ethiopia. PLoS Negl Trop Dis 2013 Nov; 7 (11):e2543. doi: <u>10.1371/journal.pntd.0002543</u> PMID: <u>24244778</u>
- Bashaye S, Nombela N, Argaw D, Mulugeta A, Herrero M, Nieto J, et al. Risk factors for visceral leishmaniasis in a new epidemic site in Amhara Region, Ethiopia. Am J Trop Med Hyg 2009 Jul; 81(1):34–9. PMID: 19556563
- Yared S, Deribe K, Gebreselassie A, Lemma W, Akililu E, Kirstein OD, et al. Risk factors of visceral leishmaniasis: a case control study in north-western Ethiopia. Parasit Vectors 2014 Oct 14; 7(1):470.
- Ryan JR, Mbui J, Rashid JR, Wasunna MK, Kirigi G, Magiri C, et al. Spatial clustering and epidemiological aspects of visceral leishmaniasis in two endemic villages, Baringo District, Kenya. Am J Trop Med Hyg 2006 Feb; 74(2):308–17. PMID: <u>16474089</u>
- Kolaczinski JH, Reithinger R, Worku DT, Ocheng A, Kasimiro J, Kabatereine N, et al. Risk factors of visceral leishmaniasis in East Africa: a case-control study in Pokot territory of Kenya and Uganda. Int J Epidemiol 2008 Apr; 37(2):344–52. doi: <u>10.1093/ije/dym275</u> PMID: <u>18184669</u>
- Elnaiem DE, Mukhawi AM, Hassan MM, Osman ME, Osman OF, Abdeen MS, et al. Factors affecting variations in exposure to infections by Leishmania donovani in eastern Sudan. East Mediterr Health J 2003 Jul; 9(4):827–36. PMID: 15748079
- Bucheton B, Kheir MM, El-Safi SH, Hammad A, Mergani A, Mary C, et al. The interplay between environmental and host factors during an outbreak of visceral leishmaniasis in eastern Sudan. Microbes Infect 2002 Nov; 4(14):1449–57. PMID: 12475635
- Ritmeijer K, Davies C, van ZR, Wang SJ, Schorscher J, Dongu'du SI, et al. Evaluation of a mass distribution programme for fine-mesh impregnated bednets against visceral leishmaniasis in eastern Sudan. Trop Med Int Health 2007 Mar; 12(3):404–14. PMID: <u>17313512</u>
- Mueller YK, Nackers F, Ahmed KA, Boelaert M, Djoumessi JC, Eltigani R, et al. Burden of visceral leishmaniasis in villages of eastern Gedaref State, Sudan: an exhaustive cross-sectional survey. PLoS Negl Trop Dis 2012; 6(11):e1872. doi: 10.1371/journal.pntd.0001872 PMID: 23133683
- Ritmeijer K, Melaku Y, Mueller M, Kipngetich S, O'keeffe C, Davidson RN. Evaluation of a new recombinant K39 rapid diagnostic test for Sudanese visceral leishmaniasis. Am J Trop Med Hyg 2006 Jan; 74 (1):76–80. PMID: <u>16407349</u>
- Chappuis F, Rijal S, Soto A, Menten J, Boelaert M. A meta-analysis of the diagnostic performance of the direct agglutination test and rK39 dipstick for visceral leishmaniasis. BMJ 2006 Oct 7; 333 (7571):723. PMID: <u>16882683</u>
- Meredith SE, Kroon NC, Sondorp E, Seaman J, Goris MG, van Ingen CW, et al. Leish-KIT, a stable direct agglutination test based on freeze-dried antigen for serodiagnosis of visceral leishmaniasis. J Clin Microbiol 1995 Jul; 33(7):1742–5. PMID: <u>7665640</u>
- Babiker ZO, Davidson R, Mazinda C, Kipngetich S, Ritmeijer K. Utility of lymph node aspiration in the diagnosis of visceral leishmaniasis in Sudan. Am J Trop Med Hyg 2007 Apr; 76(4):689–93. PMID: <u>17426171</u>
- Salih NA, van GJ, Chappuis F, Antierens A, Mumina A, Hammam O, et al. Liposomal amphotericin B for complicated visceral leishmaniasis (kala-azar) in eastern Sudan: how effective is treatment for this neglected disease? Trop Med Int Health 2014 Feb; 19(2):146–52. doi: <u>10.1111/tmi.12238</u> PMID: 24433217
- Perry D, Dixon K, Garlapati R, Gendernalik A, Poche D, Poche R. Visceral leishmaniasis prevalence and associated risk factors in the saran district of Bihar, India, from 2009 to July of 2011. Am J Trop Med Hyg 2013 Apr; 88(4):778–84. doi: <u>10.4269/ajtmh.12-0442</u> PMID: <u>23382167</u>

- Custodio E, Gadisa E, Sordo L, Cruz I, Moreno J, Nieto J, et al. Factors associated with Leishmania asymptomatic infection: results from a cross-sectional survey in highland northern Ethiopia. PLoS Negl Trop Dis 2012; 6(9):e1813. doi: <u>10.1371/journal.pntd.0001813</u> PMID: <u>23029576</u>
- Matlashewski G, Arana B, Kroeger A, Be-Nazir A, Mondal D, Nabi SG, et al. Research priorities for elimination of visceral leishmaniasis. Lancet Glob Health 2014 Dec; 2(12):e683–e684. doi: <u>10.1016/</u> <u>S2214-109X(14)70318-3</u> PMID: <u>25433617</u>
- Alvar J, Yactayo S, Bern C. Leishmaniasis and poverty. Trends Parasitol 2006 Dec; 22(12):552–7. PMID: <u>17023215</u>
- Wilson AL, Dhiman RC, Kitron U, Scott TW, van den Berg H, Lindsay SW. Benefit of insecticide-treated nets, curtains and screening on vector borne diseases, excluding malaria: a systematic review and meta-analysis. PLoS Negl Trop Dis 2014 Oct; 8(10):e3228. doi: <u>10.1371/journal.pntd.0003228</u> PMID: <u>25299481</u>
- Moncaz A, Kirstein O, Gebresellassie A, Lemma W, Yared S, Gebre-Michael T, et al. Characterization of breeding sites of Phlebotomus orientalis—the vector of visceral leishmaniasis in northwestern Ethiopia. Acta Trop 2014 Nov; 139:5–14. doi: <u>10.1016/j.actatropica.2014.06.013</u> PMID: <u>25004440</u>
- Lemma W, Tekie H, Balkew M, Gebre-Michael T, Warburg A, Hailu A. Population dynamics and habitat preferences of Phlebotomus orientalis in extra-domestic habitats of Kafta Humera lowlands—kala azar endemic areas in Northwest Ethiopia. Parasit Vectors 2014; 7:359. doi: <u>10.1186/1756-3305-7-359</u> PMID: 25098264
- Baleela R, Llewellyn MS, Fitzpatrick S, Kuhls K, Schonian G, Miles MA, et al. Leishmania donovani populations in Eastern Sudan: temporal structuring and a link between human and canine transmission. Parasit Vectors 2014 Nov 20; 7(1):496.
- Dereure J, El-Safi SH, Bucheton B, Boni M, Kheir MM, Davoust B, et al. Visceral leishmaniasis in eastern Sudan: parasite identification in humans and dogs; host-parasite relationships. Microbes Infect 2003 Oct; 5(12):1103–8. PMID: 14554251
- Hassan MM, Osman OF, El-Raba'a FM, Schallig HD, Elnaiem DE. Role of the domestic dog as a reservoir host of Leishmania donovani in eastern Sudan. Parasit Vectors 2009; 2(1):26. doi: <u>10.1186/1756-3305-2-26</u> PMID: <u>19534802</u>
- Gebre-Michael T, Balkew M, Berhe N, Hailu A, Mekonnen Y. Further studies on the phlebotomine sandflies of the kala-azar endemic lowlands of Humera-Metema (north-west Ethiopia) with observations on their natural blood meal sources. Parasit Vectors 2010; 3(1):6. doi: <u>10.1186/1756-3305-3-6</u> PMID: <u>20181077</u>